---
displayed_sidebar: jsClientSidebar
---

import ParameterTypes from "@site/src/components/ParameterTypes"

# JS Client Reference

## Modules

- [internal](modules/internal.mdx)

## Classes

- [AddressesResource](classes/AddressesResource.mdx)
- [Admin](classes/Admin.mdx)
- [AdminAuthResource](classes/AdminAuthResource.mdx)
- [AdminBatchJobsResource](classes/AdminBatchJobsResource.mdx)
- [AdminCollectionsResource](classes/AdminCollectionsResource.mdx)
- [AdminCurrenciesResource](classes/AdminCurrenciesResource.mdx)
- [AdminCustomResource](classes/AdminCustomResource.mdx)
- [AdminCustomerGroupsResource](classes/AdminCustomerGroupsResource.mdx)
- [AdminCustomersResource](classes/AdminCustomersResource.mdx)
- [AdminDiscountsResource](classes/AdminDiscountsResource.mdx)
- [AdminDraftOrdersResource](classes/AdminDraftOrdersResource.mdx)
- [AdminGiftCardsResource](classes/AdminGiftCardsResource.mdx)
- [AdminInventoryItemsResource](classes/AdminInventoryItemsResource.mdx)
- [AdminInvitesResource](classes/AdminInvitesResource.mdx)
- [AdminNotesResource](classes/AdminNotesResource.mdx)
- [AdminNotificationsResource](classes/AdminNotificationsResource.mdx)
- [AdminOrderEditsResource](classes/AdminOrderEditsResource.mdx)
- [AdminOrdersResource](classes/AdminOrdersResource.mdx)
- [AdminPaymentCollectionsResource](classes/AdminPaymentCollectionsResource.mdx)
- [AdminPaymentsResource](classes/AdminPaymentsResource.mdx)
- [AdminPriceListResource](classes/AdminPriceListResource.mdx)
- [AdminProductCategoriesResource](classes/AdminProductCategoriesResource.mdx)
- [AdminProductTagsResource](classes/AdminProductTagsResource.mdx)
- [AdminProductTypesResource](classes/AdminProductTypesResource.mdx)
- [AdminProductsResource](classes/AdminProductsResource.mdx)
- [AdminPublishableApiKeyResource](classes/AdminPublishableApiKeyResource.mdx)
- [AdminRegionsResource](classes/AdminRegionsResource.mdx)
- [AdminReservationsResource](classes/AdminReservationsResource.mdx)
- [AdminReturnReasonsResource](classes/AdminReturnReasonsResource.mdx)
- [AdminReturnsResource](classes/AdminReturnsResource.mdx)
- [AdminSalesChannelsResource](classes/AdminSalesChannelsResource.mdx)
- [AdminShippingOptionsResource](classes/AdminShippingOptionsResource.mdx)
- [AdminShippingProfilesResource](classes/AdminShippingProfilesResource.mdx)
- [AdminStockLocationsResource](classes/AdminStockLocationsResource.mdx)
- [AdminStoresResource](classes/AdminStoresResource.mdx)
- [AdminSwapsResource](classes/AdminSwapsResource.mdx)
- [AdminTaxRatesResource](classes/AdminTaxRatesResource.mdx)
- [AdminUploadsResource](classes/AdminUploadsResource.mdx)
- [AdminUsersResource](classes/AdminUsersResource.mdx)
- [AdminVariantsResource](classes/AdminVariantsResource.mdx)
- [AuthResource](classes/AuthResource.mdx)
- [CartsResource](classes/CartsResource.mdx)
- [CollectionsResource](classes/CollectionsResource.mdx)
- [CustomersResource](classes/CustomersResource.mdx)
- [GiftCardsResource](classes/GiftCardsResource.mdx)
- [LineItemsResource](classes/LineItemsResource.mdx)
- [OrderEditsResource](classes/OrderEditsResource.mdx)
- [OrdersResource](classes/OrdersResource.mdx)
- [PaymentCollectionsResource](classes/PaymentCollectionsResource.mdx)
- [PaymentMethodsResource](classes/PaymentMethodsResource.mdx)
- [ProductCategoriesResource](classes/ProductCategoriesResource.mdx)
- [ProductTagsResource](classes/ProductTagsResource.mdx)
- [ProductTypesResource](classes/ProductTypesResource.mdx)
- [ProductVariantsResource](classes/ProductVariantsResource.mdx)
- [ProductsResource](classes/ProductsResource.mdx)
- [RegionsResource](classes/RegionsResource.mdx)
- [ReturnReasonsResource](classes/ReturnReasonsResource.mdx)
- [ReturnsResource](classes/ReturnsResource.mdx)
- [ShippingOptionsResource](classes/ShippingOptionsResource.mdx)
- [SwapsResource](classes/SwapsResource.mdx)
